**Author's response to reviews**

**Title:** Cost effectiveness of preventing falls and improving mobility in people with Parkinson disease: Protocol for an economic evaluation alongside a clinical trial

**Authors:**

Jennifer J Watts (jenny.watts@buseco.monash.edu.au)
Jennifer McGinley (jennifer.mcginley@mcri.edu.au)
Frances Huxham (fhuxham@optusnet.com.au)
Hylton B Menz (h.menz@latrobe.edu.au)
Robert Iansek (robert.iansek@med.monash.edu.au)
Anna T Murphy (AnnaT.Murphy@southernhealth.org.au)
Emma Waller (emma.waller@buseco.monash.edu.au)
Meg Morris E Morris (mmorris@unimelb.edu.au)

**Version:** 2  **Date:** 15 September 2008

**Author's response to reviews:**

Ethics approval sub-section included under the method section of the manuscript.